Contents lists available at ScienceDirect

### **Ecological Economics**

journal homepage: www.elsevier.com/locate/ecolecon

## Integrating coordination mechanisms in the sustainability assessment of agri-food chains: From a structured literature review to a comprehensive framework

Carlos Moreno-Miranda<sup>a,b,\*</sup>, Liesbeth Dries<sup>a</sup>

<sup>a</sup> Agricultural Economics and Rural Policy Group, Wageningen University & Research, Hollandseweg 1, 6706, KN, the Netherlands
 <sup>b</sup> Faculty of Food Engineering and Biotechnology, Technical University of Ambato, Av. Los Chasquis y Río Payamino, 180104, Ecuador

### ARTICLE INFO

Keywords: Sustainability Performance Measurement Integrated Assessment Vertical and Horizontal Coordination Chain Configuration

### ABSTRACT

The assessment of sustainability at the level of the agri-food supply chain is complex. Achieving sustainability in agri-food chains requires coordinated interaction between chain actors. The aspect of coordination is rarely included in the assessment of sustainability and current assessment methodologies do not allow conclusions about the linkage between coordination and sustainability. This paper analyzes the state of the art in the assessment of agri-food supply chain sustainability based on a structured literature review. Following the structured review that includes category analysis and content analysis, we develop a comprehensive sustainability assessment framework for agri-food chains across multiple stages. The novelty of the framework is to incorporate the dimension of coordination across chain stages as a critical sustainability by incorporating the role of coordination across chain stages and its relationship with economic, environmental, and social performance. This essential relationship between coordination and sustainability offers several areas of interest for future research. The study also contributes to practice by providing scholars, chain actors, and policymakers with directions for improving sustainable strategies.

### 1. Introduction

The sustainability of agri-food supply chains is high on international and domestic policy agendas for corporations, governments, NGOs, academia, and societies. The issues commonly discussed are climate change (Godde et al., 2021), loss of biodiversity (Ivanov and Dolgui, 2020), and their linkages to agricultural production (Gouda and Saranga, 2018). Stakeholders have claimed that current actions are insufficient to achieve the sustainability of the supply chains (Kugelberg et al., 2021). The common issues are food waste (Vanlauwe et al., 2019), water contamination (Mihai and Ingrao, 2018), the inefficiency of energy use (Wang et al., 2020), and livelihoods deterioration (Huss et al., 2021). Concerns also exist over the increasingly imbalanced distribution of benefits across supply chain actors (Barbosa, 2021; Mani et al., 2020). These observations show that sustainability encompasses several dimensions (Kilelu et al., 2017; Um and Kim, 2019) and requires a transdisciplinary approach to tackle this priority (Green et al., 2020; Melkonyan et al., 2020).

Several authors assert sustainability in agri-food chains requires coordinated interaction between chain actors (Estevez et al., 2018; Kornher and Kalkuhl, 2019). Coordination between supply chain actors/ stages occurs through mechanisms such as direction (Bijman et al., 2006), planning methods (Macdonald, 2020), logistic arrangements (Paciarotti and Torregiani, 2021), incentives (Ward et al., 2016), mutual adjustment (Rydberg and Haden, 2006), and information transfer (Gerbens-Leenes et al., 2003), and it can involve formal or informal arrangements (Borgen and Hegrenes, 2005). Through such mechanisms, chain actors may achieve a better sustainability performance (Msaddak et al., 2017), for instance, by reducing food losses and contamination being more efficient and improving smallholders' position in the chain (Gimenez and Tachizawa, 2012; Zhong et al., 2018). The few existing studies that include coordination in the sustainability assessment focus primarily on cooperation for managing conflicts (Gonzalez-Perez and Gutierrez-Viana, 2012) and do not provide conclusions on the relationship between coordination and sustainability (Mausch et al., 2020). This is the knowledge gap that the current article seeks to address.

\* Corresponding author at: Agricultural Economics and Rural Policy Group, Wageningen University & Research, Hollandseweg 1, 6706, KN, the Netherlands. *E-mail addresses:* carlos.morenomiranda@wur.nl (C. Moreno-Miranda), liesbeth.dries@wur.nl (L. Dries).

https://doi.org/10.1016/j.ecolecon.2021.107265

Received 26 March 2021; Received in revised form 16 October 2021; Accepted 17 October 2021 Available online 3 November 2021 0921-8009/© 2021 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).







Many reviews have provided guidelines and methods to estimate the sustainability performance of agri-food supply chains. These include Life-Cycle-Assessment (LCA), Balanced Score Cards (BSC), and Data Envelopment Analysis (DEA). Often these methods focus on only one chain stage or a dyad (Martins and Pato, 2019; Peterson, 2010; Fan et al., 2016) and do not simultaneously analyze all sustainability dimensions (Hoekstra et al., 2019; Kozlowski et al., 2015). Their limitations also are evidenced in their abstract approaches for measuring with incomplete interpretations (Charles et al., 2006) of the outcomes (Camargo et al., 2019; Shafiq et al., 2014), and the thoughtlessness of the potential of the coordination between stages to the sustainability of the supply chains (Beske and Seuring, 2014; Dania et al., 2018). More research on the sustainability assessment of agri-food chains is supported by Zhu et al. (2017) and Chen et al. (2017), while Ansari and Kant (2017) call for research on the integrated measurement of supply chain sustainability.

To shed light on the identified gaps, we will analyze the state of art in assessing agri-food supply chain sustainability based on a structured literature review. We consider the supply chain a suitable unit of analysis and include coordination between vertical linkages of chain stages as a crucial aspect of the assessment. Although we consider horizontal connections between actors at a specific chain stage, we do not consider the supply network as the unit of analysis because this would encompass clusters of chains and require more complex analysis that misled our goal, the supply chain level. Following the structured review, we develop a comprehensive sustainability assessment framework for agrifood chains across multiple stages. The novelty of the framework is to incorporate the dimension of coordination across chain stages as a critical sustainability dimension.

The following research questions will guide the structured literature review to determine a comprehensive framework for the assessment of agri-food chain sustainability:

- a) What is the state of art in assessing agri-food chain sustainability, and which sustainability dimensions and assessment methods are reviewed in the existing literature?
- b) Which chain-stages are included in the existing assessment approaches, and how is the connection across chain stages taken into account?
- c) What are the key elements and indicators to be included in a comprehensive sustainability assessment for agri-food supply chains?

The research is based on a structured literature review and content analysis of 109 peer-reviewed articles published from 1997 to 2018 on agri-food supply chains. Criteria in the analysis include the agri-food industry, sustainability dimensions, chain levels, assessment methods, and theoretical perspectives.

To ensure relevance and feasibility, the study tested the assessment framework in Ecuador's fruit and vegetable chains since this country facilitated the logistics for the framework testing. To this end, we organized workshops with stakeholders from locations where there exists a prominent agro-alimentary sector. These locations allow easy access to a diversity of agri-food chain actors because their distribution centers are frequented by 25,000 (daily, as a minimum) local and foreign producers, intermediaries, wholesalers, transport companies, and consumers. Of interest to the study, these locations are ahead of other locations in adopting sustainability practices. Primary evidence of this is the launch of the agenda sustainable agriculture 2017 to reinforce the locations' sustainability.

The research makes several contributions to the literature on the sustainability assessment of agri-food supply chains. First, we provide a comprehensive overview of a vast number of studies on sustainability assessment approaches in agri-food supply chains. Second, we identify different types of coordination mechanisms in the evaluation of chain sustainability. Third, we discuss the limitations of existing sustainability assessments. Fourth, we present a novel conceptual framework for

assessing agri-food supply chains' sustainability performance, and, fifth, we test the relevance and feasibility of this framework using a participatory approach.

The remainder of the article is structured as follows. Section 2 provides key concepts in the research. Section 3 presents the methodology for the structured literature review and the framework testing. Section 4 shows the content analysis findings and develops the foundations of the conceptual framework for assessing the sustainability performance of agri-food supply chains. Section 5 presents the outcomes of the framework testing. Section 6 discusses the relevance of the findings in the context of the literature on agri-food chain sustainability assessment, the implications for practice, and limitations of the research. Section 7 concludes the article.

### 2. Conceptual Background

### 2.1. Sustainability Performance

Sustainability performance relates to results and achievements in the domain of sustainability. Sustainability incorporates three core dimensions, namely, economic, environmental, and social. Considering the supply chain context, we define sustainability performance as the economic, social, and ecological potential to be met simultaneously for the supply chain to maintain supply operations over time. Environmental performance considers efficiency in resource utilization (Schanes et al., 2018), recycling (Boesen et al., 2019), and reduction of pollution (Rao and Holt, 2005). Social performance considers human rights and impact on local communities (Yawar and Seuring, 2017). Economic performance operationalizes profitability, and accounting-based metrics (Golicic and Smith, 2013).

### 2.2. Sustainability Assessment

Sustainability assessment involves an attempt to evaluate sustainability performance. Qualitative and quantitative tools and indicators are crucial elements of an assessment design (Von Wirén-Lehr, 2001). The vital aspects of its operationalization are stakeholder engagement and contextualization. The former involves key actors (Gregory et al., 2020; Moreno-Miranda et al., 2020), and the latter bridges the design with technical needs (Abildtrup et al., 2006). We define sustainability assessment in the supply chain context as the method to gather, analyze, and transfer economic, social, and ecological information and comprehend the supply chain context. Examples of assessment methods are hierarchy evaluation (Forman and Gass, 2001) and product lifecycle analysis (Baudino et al., 2017).

### 2.3. Coordination and Coordination Mechanisms

The management of interdependencies (relationships) is done through coordination mechanisms (Koberg and Longoni, 2019; Tachizawa and Wong, 2014). Coordination is the creation, maintenance, and transformation of relationships between actors (Gimenez and Tachizawa, 2012; Malone and Crowston, 1994; Raynold, 2004). Coordination mechanisms are practices (Huang et al., 2016) and processes (Jiao et al., 2021) through which chain actors are governed (Carter and Easton, 2011). These mechanisms can be horizontal or vertical (Moreno-Miranda et al., 2019). Vertical mechanisms are practices to manage interdependencies across supply chain stages (Herzog et al., 2018; Peterson et al., 2001). Horizontal mechanisms indicate the relationship between peers positioned at the same chain level (Sting and Loch, 2016). Examples of these mechanisms include mutual adjustment (Douma and Schreuder, 2017), standards (Dasgupta, 1996), and direct supervision (Slange et al., 2008).

Coordination mechanisms play a crucial role in the sustainability performance of agri-food chains because they can reduce uncertainty and complexity of transactions. Unrelieved tensions may lead to significant transaction costs and affect sustainability (Gereffi et al., 2005; Vroegindewey et al., 2018). According to Formentini and Taticchi (2016), coordination mechanisms are elements adopted by actors to coordinate actions, optimize economic utility, promote social welfare and amplify ecological resilience.

### 3. Methodology

This paper will first perform a structured review of the literature to develop a comprehensive framework for assessing the sustainability of agri-food supply chains. Second, it will perform the framework testing to estimate the framework's relevance and feasibility.

### 3.1. Structured Literature Review

A structured literature review (SLR) allows for collecting and analyzing a large amount of evidence transparently and reliably (Sutton et al., 1999). The SLR brings the reader up-to-date with the current literature and sets future research directions (Cronin et al., 2008; Webster and Watson, 2002). We consult review examples performed by Farooque et al. (2019) and Gunawan et al. (2021); and we follow the methodology proposed by Seuring and Gold (2012) using a two-stage process: 1) material collection and 2) a structured review consisting of category identification and content analysis.

### 3.1.1. Material Collection

In the material collection stage, we define the scope of the material to be analyzed. To ensure that relevant studies are screened in the review, we limit the search based on the following considerations:

- We account for the research domains of agricultural economics, food systems, supply chain research, agricultural sustainability, ecological economics, agricultural research, and development economics for proper interpretation and validity of the findings.
- 2. The analysis considers peer-reviewed articles and conference papers published in English since it is the language of choice for many international scholarly journals.
- 3. The search deploys a combination of key search terms to select relevant papers. For specifying the context, we use terms related to "agri-food" OR "agr\*" OR "food\*" OR "farm\*" to cover agroprocessing, agribusiness, food-processing, and farm-related settings. For specifying our interest in sustainability, we employ "sustain\*" OR "environment\*" OR "fair\*" OR "feasib\*" OR "life\*" OR "renewab\*" OR "social\*" OR "coordinat\*". For capturing the supply chain dimension, we use terms such as "network" OR "market chain" OR "supply chain" OR "value chain". For covering assessments in terms of performance, we use "perform\*" OR "efficien\*" OR "product\*" OR "profit\*" OR "manage\*". We combine these key terms to generate a series of strings, e.g. ("agri-food" OR "food\*" OR "farm\*") AND ("sustain\*" OR "environment\*" OR "feasib\*"OR "life\*" OR "social\*" OR "collaborat\*") AND ("network"OR "supply chain"OR
- 4. We target documents written by experts in the field and published in the period ranging from 1997 to 2018. The search was performed in multiple databases, including CAB Abstract via Ovid (www.ovid. com), AgEcon search (ageconsearch.umn.edu), Web of Science (apps.webofknowledge.com), WUR library collection (www.wur. nl/Library.htm), and Scopus (www.scopus.com) since they provide powerful tools for optimizing results.
- 5. Criteria considered to exclude studies were the following:
  - a. Publications aimed at chemistry, medicine, genetics, nanotechnology, computing, and physics because they differ from our socio-economic target.
  - b. Publications focused on pure ecological issues, e.g., climate cycles and plant physiology. The interest in physical phenomena differs from our objective.

- c. Publications on product marketing and logistics because focus mainly on sales, and we exclude novel food chanis, e.g., insects, because of their emphasis on diets replacement.
- d. Publications related to supply chains outside of the agri-food sector such as chemicals, automotive, clothing, electronics, among others, because of the different nature of their exchanged goods.
- 6. After exclusion based on the criteria in step 5 and the exclusion of duplicates, 278 papers remain. We then proceed by screening abstracts to assess if they fit our research questions. We do not consider articles that emphasize personal opinions, legal conclusions, purely historical papers and lack meta-commentary to facilitate the abstracts screening. This results in 99 studies to be used for analysis.
- 7. Next, we also use references from these 99 studies to identify additional relevant papers. For example, from the article by (Pullman et al., 2010), which was retrieved from the search, we identify (Aramyan et al., 2006) as a relevant article not captured by our keyword search.
- 8. Finally, we account for 109 documents in the final sample. Fig. 1 presents the contribution of each database and steps in the material collection process.

Fig. 2 shows that most of the articles in our sample are published in the last seven years. This is in line with the growing interest of scholars in the sustainability evaluation of agri-food supply chains.

### 3.1.2. Structured Literature Review

The description of the structured review approach is shown in Fig. 3. The structured review follows two steps performed iteratively: 1) category analysis; 2) content analysis. Category analysis is based on inductive, deductive, or context review. The inductive review draws conclusions based on abstracts (Thomas, 2006). The deductive review outlines general principles (Manna and Waldinger, 1986). Further, context review incorporates insights from circumstances or events (Small, 1980).

We start the category review by reading each article's abstract in our sample to determine the agri-food supply chains and sustainability indicators (inductive phase). Next, we review the conceptual frameworks to elicit relevant theories and derive insights into the foundations of sustainability assessment (deductive review). For instance, Carter and Rogers (2008) use the resource dependence theory. Others have used systems theory (Holmberg, 2000), stakeholder theory (Mutebi Kalibwani et al., 2018; Neves, 2010), transaction cost theory (Gereffi, 1994), or the value-chain approach (Porter, 1985). We then complement the analysis by reviewing context features, such as transitions. Following the category analysis, we conduct a content analysis focused on chain configuration and coordination mechanisms. The aspect of chain configuration was present in the nested subsample of 27 documents, and 12 out of these 27 articles address coordination mechanisms. This analysis uses frequency, trend, and theme identification. The purpose of content analysis is to identify the chain-level and coordination elements to be part of the final framework.

### 3.2. Framework Testing

To test and operationalize the conceptual framework, a workshop was organized with actors and stakeholders involved in fruit and vegetable agri-food chains in Ecuador. The workshop was conducted in Spanish, the participants' native language. The event took place in December 2019 and was attended by 45 participants: representatives from the production (16), processing (8) and distribution (10) stages of the supply chains, from academia (6) and governmental institutions (5). The production-stage representatives included smallholders and leaders of farmers' associations. The heads of public distribution centers, intermediaries, and food processors represented downstream stages. The ministry of agriculture, the public water board, the agricultural



Fig. 1. Representation of the material collection process. Source: Authors' own representation.



**Fig. 2.** Distribution of sample articles by year. Source: Authors' own representation.

parliament, and universities represent governmental institutions and academia.

The workshop's primary purpose was to identify the conceptual framework's relevance and feasibility in a real-life context. The participants assessed both aspects to prevent the sustainability assessment becomes dominated by few stakeholders' views. The workshop started with a description of the framework's foundations and implications. The second phase of the workshop used a questionnaire based on the sustainability indicators derived from the structured literature review. Cronbach's alpha coefficient validated the questionnaire, and three experts checked the wording in the questionnaire. The survey was then pilot-tested<sup>1</sup> with five interviewees, who assisted further in eliminating ambiguous items. The final questionnaire includes 16 environmental, 11 social, 13 economic, and 18 coordination indicators.

<sup>&</sup>lt;sup>1</sup> First, ten respondents validated each indicator in terms of consistency (understandability). The assigned scores followed a scale from 1 to 10. We estimate the Cronbach alpha coefficient by employing the formula CA = (K/K-1) x ((SUM variance of question i)/(variance of observed scores of individuals)), where K = number of questions. Cronbach alpha values range from 0 to 1.0 points. Scores between 0.60 and 0.70 mean acceptable reliability. We obtained coefficients of over 0.80.



**Fig. 3.** Representation of the structured revision process. Source: Authors' own representation.

During the workshop, participants assessed the relevance and feasibility of each indicator. Relevance refers to the degree of the practical usefulness in a real-life context. Feasibility refers to the ease of collecting the information. We estimate the relevance by using absolute frequencies of a 5-point Likert scale (1 extremely irrelevant, 2 irrelevant, 3 neutral, 4 relevant and 5 extremely relevant) per indicator and by using Eq. (1) as follows:

$$RR = \frac{(5^*n_5) + (4^*n_4) + (3^*n_3) + (2^*n_2) + (1^*n_1)}{A^*N}$$
(1)

Note: RR = relative relevance;  $n_5 =$  number of respondents indicating extremely relevant;  $n_4 =$  number of respondents indicating relevant;  $n_3 =$  number of respondents indicating irrelevant;  $n_1 =$  number of respondents indicating extremely irrelevant; A = highest score (5); N = total number of respondents.

Outcomes of the relative relevance analysis are then used to create radar plots. Five experts from universities and public institutions and five stakeholder group representatives assessed the indicators' feasibility using a qualitative scale (low, medium, and high). The feasibility criteria that were used are applicability, execution, and data availability. Applicability refers to the appropriateness, execution refers to the measurement; and data availability involves access to primary data. Absolute frequencies of expert responses were used to generate bar charts.

### 4. Findings of the Content Analysis and Conceptual Framework

Fig. 4 shows that energy and engineering science contributes 26% (28 papers) of the total sample of 109 articles. This shows a clear interest in the linkage between energy efficiency and the performance of agrifood chains. Management and environmental science contribute 18% (20 articles) to the sample, and 11% (12 articles) come from the social sciences. In the following sections, we look more closely at the sustainability dimensions, the indicators and criteria, and the methodologies used in the sample articles (see Appendix A for the list of articles that provided guidelines to estimate sustainability). We then turn to the content analysis.

### 4.1. Sustainability Dimensions and Indicators

Table 1 categorizes the sample's articles by sustainability dimension and agri-food chain. We identify articles with single and joint assessments of the three sustainability dimensions (e.g., Van Der Werf and Petit, 2002), namely economic, social, and environmental as single assessments and economic-environmental, socio-economic, socioenvironmental, and triple-bottom-line (3BL) as joint assessments (e.g.,



Fig. 4. Distribution of the articles by research subjects.

Source: Authors' own representation based on sample articles in Appendix A. Note: SS social sciences; MEDS mathematics, economics, and decision sciences; ABS agriculture and biological sciences; BM management; ES environmental sciences; EES energy and engineering sciences.

Chaparro and Calle, 2017; Longo et al., 2017; Thomassen et al., 2009). The organizational dimension is identified in articles across the different categories. In the subgroup of single assessments, the environmental dimension is most frequent in scientific contributions (34%). The economic and the social perspective have received limited coverage (6% and 9% of all manuscripts, respectively). Joint assessments contribute between 11 and 15 articles to the sample. Only 11% (12 articles) tackle the organizational aspect of sustainability, of which 4 articles are single assessments, and 8 articles contribute to the socio-economic subgroup. The identified organizational aspects include supply chain contracts, interdependencies, vertical and horizontal collaboration, cooperation, and transaction costs.

More than 40% of the sample articles examine foodstuffs in general without product specification. Fruits and vegetables and cereals are the most common chains assessed. Dairy, meat, and coffee chains have also received attention in the relevant literature.

### Table 1

Distribution of studies by agri-food chain and sustainability dimension (%).

|                 |                    | Sustainabili | ty dimensio | on            |                        |                |                     |       |         |
|-----------------|--------------------|--------------|-------------|---------------|------------------------|----------------|---------------------|-------|---------|
|                 |                    | Single       |             |               | Joint                  |                |                     |       |         |
|                 |                    | Economic     | Social      | Environmental | Economic-environmental | Socio-economic | Socio-environmental | 3BL   | Total % |
| Agri-food chain | Oil                |              |             |               |                        | 2 (2)          |                     |       | 2       |
|                 | Cereals            | 1 (1)        | 1(1)        | 6 (6)         | 4 (4)                  | 2 (2)*         | 1 (1)               | 2 (2) | 16      |
|                 | Honey              |              |             | 1 (1)         |                        |                |                     |       | 1       |
|                 | Coffee             | 1 (1)*       |             |               | 1 (1)                  |                | 4 (4)               |       | 6       |
|                 | Egg-hen            |              | 1(1)        | 2 (2)         |                        |                |                     |       | 3       |
|                 | Dairy              |              | 1(1)        | 1 (1)         | 1 (1)                  | 1 (1)*         | 1 (1)               | 2 (2) | 6       |
|                 | Sugar              |              |             | 1 (1)         |                        |                |                     |       | 1       |
|                 | Meat products      |              |             | 2 (2)         | 1 (1)                  |                |                     |       | 3       |
|                 | Fruit & Vegetables |              | 1(1)        | 15 (16)       | 1 (1)                  | 4 (4)*         |                     | 2 (2) | 22      |
|                 | Wine               |              |             |               |                        |                | 1 (1)               |       | 1       |
|                 | Foodstuff          | 5 (5)*       | 6 (6)*      | 7 (8)*        | 6 (7)                  | 6 (6)*         | 6 (6)*              | 6 (6) | 40      |
|                 | Total %            | 6            | 9           | 34            | 14                     | 14             | 12                  | 11    | 100     |

Source: Authors' own representation based on sample articles in Appendix A.

Notes: \* identifies categories that include one or more articles covering aspects of organizational sustainability; 3BL - triple-bottom-line is a framework with a focus on all three sustainability dimensions: economic, social and environmental; xx (xx) – percentage (absolute value) of articles per subgroup.

Next, we discuss the sustainability indicators used in the sample articles. We focus on the sustainability dimensions commonly used in the literature: economic, environmental, and social. The organizational dimension will be discussed in Section 4.3.2. Tables 2a to 2c provide an overview. The economic indicators refer to elements of productivity, profit, and prices. The environmental indicators are related to resource use and ecological consequences. Social indicators refer to the worker, the value chain actor, and the local community. We elicit 13, 16, and 11 indicators for economic, environmental, and social dimensions, respectively. Economic indicators measure the financial situation of value chains (Diener and Suh, 1997). Environmental indicators are metrics of ecological conditions (e.g., energy use, land conservation) (Niemeijer and de Groot, 2008; Schmidt et al., 2014; Tost et al., 2018) to assess environmental objectives, support stakeholders in their decisionmaking processes and detect trends. Social indicators describe features and operations of populations in an aggregate way (Panagiotakopoulos et al., 2016; Xie, 2016). They evaluate the degree of wellbeing goals achieving (e.g., health, education, and housing).

### 4.2. Sustainability Assessment Methods

Table 3 categorizes the sample articles based on the assessment method used. We identify twelve methods and group them into four categories. The integrated assessment (IA) method is the most common in articles (52%). This method stresses the appropriateness of indicators and uses qualitative and quantitative tools (Hasna, 2009; Moustier and Leplaideur, 1999; Wustenberghs et al., 2015). The integrated assessment method allows applying frameworks and tools (e.g., interviews, chain schemes) to cover sustainability dimensions jointly (see the red dashed box in Table 3). A drawback is that many articles that use IA remain conceptual and lack an application to real-life settings. The second most common category includes articles using life-cycle assessments (LCA) and variations (life-cycle costing, social life-cycle) (27%). This method has been mainly used to assess the environmental dimension (Engert and Baumgartner, 2016; Luo et al., 2009; Dreyer et al., 2006). Third, we identify the interdependency assessment category, which evaluates mutual relationships (e.g., economic and environmental). This includes mathematical optimization, input-output models, cross-case analysis, cluster analysis, cost-benefit analysis, and participatory design. These methods contribute 16% of articles in the sample, and almost half of them have examined economic-environmental sustainability. The fourth category includes articles using a multi-criteria approach, contains fuzzy logic, analytical hierarchy process, and data envelopment analysis (e.g., Cornelissen et al., 2001; Van Cauwenbergh et al., 2007) and contributes 5% of articles.

Fig. 5 presents the occurrence of assessment methods in articles in the period from 1997 to 2018. The integrated assessment approach have been published throughout this period. LCA approach emerged in 2008 but has taken off rapidly since 2012. Interdependency and the multi-criteria approaches have gained popularity in recent years.

So far, we have categorized the articles based on their coverage of the sustainability dimensions, the agri-food chains they focus on, and their assessment method. We have also provided an overview of the indicators. The next section will discuss the chain configurations and co-ordination mechanisms in supply chains sustainability.

### 4.3. Content Analysis

The content analysis will be based only on the articles covering the aspects of interest for developing a comprehensive conceptual framework in Section 5: i) joint sustainability dimensions and ii) the integrated assessment as the method for evaluation. The joint sustainability dimensions encompass economic, ecological, and social performance. The integrated assessment method is flexible in introducing variables, its extensive application in the literature, and its capacity to include multiple analysis tools. In total, 27 papers present these features.

### 4.3.1. Chain Configurations

The chain configurations, also called chain arrangements, are essential for determining supply chain sustainability. Chain configurations include features such as the actors, functions, and the existing linkages between stages. Fig. 6 shows seven chain configurations (and frequencies) derived from the 27 selected papers. We distinguish configurations of only one chain-stage (single, e.g., production) and multiple chain stages (multi-stage, e.g., production and distribution). The production is the most examined stage (58%). Articles focusing on the processing stage represent 3% of the sample, whereas the literature search identified no studies on the distribution stage. Articles covering multi-stage configurations (production-processing, productiondistribution, and processing-distribution) represent 24% of all contributions. About 15% of the articles consider the whole chain (cases of multinationals) in their analysis. The next section will look at the interlinkages or coordination mechanisms of the chain stages.

### 4.3.2. Coordination Mechanisms

We identify 12 articles within the full sample of 109 documents that examine coordination mechanisms as one aspect of sustainability in agri-food supply chains. Table 4 presents the vertical and horizontal indicators. Vertical indicators tackle coordination between trading partners, and horizontal indicators address coordination between

| review.    |
|------------|
| literature |
| structured |
| the        |
| from       |
| retrieved  |
| ndicators  |
| Economic i |

| Economic            |                        |                         |                              |        |       |                     |                               |                     |                    |                      |                  |          |                     |
|---------------------|------------------------|-------------------------|------------------------------|--------|-------|---------------------|-------------------------------|---------------------|--------------------|----------------------|------------------|----------|---------------------|
| Category            | Productivity           |                         |                              |        |       | Profit              |                               |                     |                    |                      |                  |          | Price               |
| Indicator           | Labour<br>productivity | Capital<br>productivity | Total factor<br>productivity | Patent | R&D   | Return on<br>assets | Return on capital<br>employed | Return on<br>equity | Return on<br>sales | Debt as<br>financing | Own<br>financing | Training | Price<br>volatility |
| Measurement<br>unit | Ratio                  | Ratio                   | Ratio                        | Level  | Level | Ratio               | Ratio                         | Ratio               | Ratio              | Level                | Level            | Level    | Ratio               |

Source: Authors' own representation based on sample articles in Appendix A.

 Table 2b

 Environmental indicators retrieved from the structured literature review.

| Environmenta  | P                 |                   |               |               |                |              |            |                 |          |               |                    |           |              |               |       |          |
|---------------|-------------------|-------------------|---------------|---------------|----------------|--------------|------------|-----------------|----------|---------------|--------------------|-----------|--------------|---------------|-------|----------|
| Category      | Resourc           | ce use            |               |               |                |              |            |                 |          |               |                    | н         | cological co | nsequences    |       |          |
| Indicator     | Water             | Irrigation water  | Water         | Organic       | Organic matter | Agrochemical | Fossil     | Crop            | Crop     | Electricity I | Deforestation Land | 0         | 202          | Acidification | Food  | Training |
|               | use               | productivity      | conservation  | matter use    | productivity   | use          | Fuel Use   | diversification | rotation | use           | consei             | rvation e | missions     | potential     | loses |          |
| Measurement   | ha/m <sup>3</sup> | kg/m <sup>3</sup> | Ratio         | ha∕kg         | Ratio          | kg/ha        | %, J, cal, | Ratio           | Ratio    | Watts/week F  | tatio Ratio        | X         | .g CO2       | g SOx         | Ratio | Level    |
| unit          |                   |                   |               |               |                |              | kWh        |                 |          |               |                    | e         | quivalent    | equivalent    |       |          |
| Source: Autho | rs' own r         | epresentation ba: | sed on sample | articles in A | ppendix A.     |              |            |                 |          |               |                    |           |              |               |       |          |

### Table 2c

Social indicators retrieved from the structured literature review.

| Social    |           |        |       |                    |                 |                  |                        |           |                         |                                |                       |
|-----------|-----------|--------|-------|--------------------|-----------------|------------------|------------------------|-----------|-------------------------|--------------------------------|-----------------------|
| Category  | Worker    |        |       |                    |                 |                  |                        |           | Value chain actor       |                                |                       |
| Indicator | Ethnicity | Gender | Age   | Education<br>level | Child<br>labour | Salary<br>ranges | Working<br>hours range | Migration | N-firm<br>concentration | Activation of social insurance | Customer satisfaction |
| unit      | Share     | Share  | Share | Share              | Share           | Range            | Range                  | Rate      | Market share            | Level                          | Score                 |

Source: Authors' own representation based on sample articles in Appendix A.

#### Table 3

Distribution of studies by assessment method and sustainability dimension (%).

|                               |                                | Sustainabili | ty dimens | ion           |                            |                    |                         |          |       |
|-------------------------------|--------------------------------|--------------|-----------|---------------|----------------------------|--------------------|-------------------------|----------|-------|
|                               |                                | Economic     | Social    | Environmental | Economic-<br>environmental | Socio-<br>economic | Socio-<br>environmental | 3BL      | Total |
| IA                            | Integrated assessment          | 2 (2)        | 6 (6)     | 8 (8)         | 9 (9)                      | 10 (10)            | 9 (9)                   | 8<br>(8) | 52    |
| LCA                           | Life-cycle-assessment<br>(LCA) |              |           | 23            | 2 (2)                      |                    | 1 (1)                   | 2<br>(2) | 27    |
| Interdependency<br>assessment | Mathematical optimization      |              |           | 1 (1)         | 1 (1)                      | 1 (1)              |                         | 1<br>(1) | 4     |
|                               | Input-output model IO          |              |           |               | 1 (1)                      | 1 (1)              |                         |          | 2     |
|                               | Cross Case analysis            | 1 (1)        | 2 (2)     | 1(1)          |                            |                    | 1 (1)                   |          | 5     |
|                               | Cluster Analysis               |              |           |               |                            | 1 (1)              |                         |          | 2     |
|                               | Cost-benefit-analysis          | 1 (1)        |           |               |                            |                    |                         |          | 1     |
|                               | Participatory design           |              | 1(1)      |               |                            |                    | 1 (1)                   |          | 2     |
| Multi-criteria approach       | Fuzzy Logic                    | 1 (1)        |           |               |                            |                    |                         |          | 1     |
|                               | Analytical Hierarchy           | 1 (1)        |           | 1 (1)         |                            |                    |                         |          | 2     |
|                               | Process (AHP)                  |              |           |               |                            |                    |                         |          |       |
|                               | Data Envelopment Analysis      |              |           |               | 1 (1)                      |                    |                         |          | 1     |
|                               | (DEA)                          |              |           |               |                            |                    |                         |          |       |
|                               | Hybrid Fuzzy-AHP-DEA           |              |           |               |                            | 1 (1)              |                         |          | 1     |
|                               | Total                          | 6            | 9         | 34            | 14                         | 14                 | 12                      | 11       | 100   |

Source: Authors' own representation based on sample articles in Appendix A.

Notes: xx (xx) – percentage (absolute value) of articles per subgroup.



□ Integrated Assessment □ Life-cycle-assessment □ Multi-criteria □ Interdependency assessment

Fig. 5. Distribution of assessment methods by year.

Source: Authors' own representation based on sample articles in Appendix A.

organization members.

# 4.4. A Conceptual Framework for Assessing Sustainability in Agri-Food Supply Chains

Assessing the sustainability of agri-food chains as a whole requires a multi-stage analysis and identification of vertical and horizontal mechanisms. The comprehensive framework presented in Fig. 7 incorporates includes the three key aspects of joint sustainability dimensions, a multistage analysis, and vertical and horizontal relationships. These three core aspects are operationalized using an integrated sustainability performance assessment methodology, with suitable indicators and analytical tools based on the chain context. The novelty and comprehensiveness of the framework lie in integrating the core dimensions of sustainability with characteristics of the supply chain, namely individual chain stages and interconnections between these stages.

### 5. Results of the Framework Testing

Figs. 8, 9, 10, 11a, and 11b show the relevance perception of the framework's indicators. Radar plots present the perception of each stakeholder's group.



**Fig. 6.** Chain-stage modes retrieved through the content analysis. Source: Authors' own representation based on sample articles. Note: xx (xx%) number of articles (share in the 27 sample articles).

### 5.1. Economic Dimension

Fig. 8 shows the relative relevance of economic indicators according to the stakeholders. All stakeholder groups put forward *price volatility*, *return on sales*, and *training level* as highly relevant indicators. On average, *research and development (R&D) and return on equity* are regarded as the least pertinent. Indicators such as *capital productivity*, *total factor productivity*, show a discrepancy of relevance perception between groups.

### 5.2. Environmental Dimension

Three out of sixteen environmental indicators are scored extremely relevant by all the groups (Fig. 9). These indicators are  $CO_2$  emissions, food losses, and training levels. Water conservation and fossil fuel use scored with high. Organic matter productivity and potential acidification indicators have significant differences in relevance perception.

### 5.3. Social Dimension

Fig. 10 shows that *working hours range* and *salary range* indicators are deemed highly relevant across stakeholders. *Ethnicity and age distribution* are scored as less relevant by the stakeholders. Indicators such as *customer satisfaction,* and *migration rate,* show a discrepancy in the relevance perception of different stakeholder groups.

### 5.4. Coordination Dimension

Fig. 11a and Fig. 11b show the relative relevance allocated by stakeholder groups to each coordination indicator. Within the vertical indicators (coordination between trading partners), stakeholders find the *interaction frequency, informal transactions*, and *written contracts* the most relevant to agri-food sustainability. Information sharing is scored as less relevant by the stakeholders. The relevance of *trust* and *distribution of bargaining power* show significant discrepancies between stakeholders.

Regarding the horizontal indicators (coordination between organization members), the stakeholders consider association and cooperative memberships as extremely relevant indicators. Indicators such as membership duration and interaction frequency within organizations also score high relevance values across groups. Trust level and informal/formal interaction frequency have the lowest relevance.

### 5.5. Feasibility Analysis

Fig. 12 shows that training, return on sales, and financing sources - own

and debt- are regarded as the most feasible economic indicators. Price volatility is assessed as medium-feasible by experts due to limited information availability. R&D is considered the least feasible indicator for an assessment across all chain levels.

Experts and stakeholder representatives indicate the *training level* as the most feasible environmental indicator (Fig. 13). *CO2 emissions, food losses,* and *water use* are reasonably feasible with current technological tools. *Land conservation* and *acidification potential* are considered not feasible in a supply-chain context.

Fig. 14 shows that experts consider salary and working hours range very feasible to assess the social aspect. Social insurance activation and migration rate are deemed workable as long as secondary data support the assessment. Customer satisfaction is seen as the most impractical indicator by experts.

Fig. 15a and Fig. 15b show the feasibility of coordination indicators. Concerning the vertical indicators, experts see *interaction frequency, information sharing*, and *informal transaction frequency* as the most feasible indicators. Indicators such as the *written contract frequency, relationship duration*, and *trust level* have medium feasibility. Bargaining power distribution is the least feasible because it demand too much effort.

Concerning the horizontal indicators, the experts consider *association* / *cooperative membership* as highly feasible indicators. Medium feasibility is assigned to the *organization frequency interaction* and *membership duration*. *Written contracts* and *informal interaction* have the lowest feasibility.

### 6. Discussion

This paper aimed to develop a comprehensive sustainability assessment framework for agri-food supply chains. Specific attention was paid to the assessment across chain stages. The coordination across chain stages and its connection with sustainability is integrated into the framework. This is the study's major contribution to the literature, which currently lacks the integration of the multi-stage element. The framework developed is tested in the context of fruit and vegetable chains. The following sections discuss the research about the existing literature, practical implications and its limitations.

# 6.1. Joint Sustainability Dimensions and the Integrated Assessment Method

The structured literature review shows that most of the existing research is focused on the environmental dimension of sustainability. This focus can be explained by the attention of agricultural stakeholders and the community for scarcity of resources (e.g., irrigation water) and

|          | ıt         | Proactiveness<br>Level                   | Score                          |
|----------|------------|--|--------------------------------|
|          | Empowermer | Social<br>Integration<br>Activities      | Relative<br>share              |
|          |            | Communication<br>level                   | Score                          |
|          |            | Tolerance<br>nce Level                   | Score                          |
|          |            | on Task<br>Compliaı                      | Ratio                          |
|          | oration    | ion Adaptatio<br>Level                   | Score                          |
|          | Collab     | ion Durati                               | q                              |
|          |            | Communicati<br>y Level                   | Score                          |
|          |            | Task<br>, Efficienc                      | Ratio                          |
|          |            | Human<br>Capital<br>y Availability       | Ratio                          |
| ntal     | ation      | Material<br>on Resources<br>Affordabilit | Ratio                          |
| Horizoi  | Cooper     | Duratio                                  | d                              |
|          |            | Device<br>Type                           | Written,<br>verbal<br>agreemen |
|          |            | Transaction<br>e Simplicity              | Perception                     |
|          |            | ct Contract<br>on Complianc              | Ratio                          |
|          | quantity   | ce Contra<br>ter Duratio                 | р г                            |
| cal      | quality,   | act Prio<br>ality Sett                   | ive Lea<br>firm                |
| Vertic   | Price,     | Contr<br>Form:                           | ıt Relati<br>share             |
| Category |            | Indicator                                | Measuremer<br>unit             |

Coordination indicators retrieved from the structured literature review

Table 4

Source: Authors' own representation based on sample articles

Ecological Economics 192 (2022) 107265

the restoration of critical impacts (Lejars et al., 2012). Govindan (2018) suggests that improving agri-food sustainability requires that stake-holders' goals cover a broad spectrum of purposes and users.

Sustainability research in recent years increasingly covers the multiple dimensions of sustainability jointly. Public and private actors increasingly call for integrated approaches to replace one-dimensional assessments. This is supported by Grimm et al. (2016), who find that firms are increasingly writing reports on economic, environmental, and social outcomes. Joint sustainability dimensions boost the integrated assessment by allowing stakeholders to critically analyze their impacts (Lee et al., 2021; Lie et al., 2012). A comprehensive framework, therefore, covers all three sustainability dimensions.

Assessment methods are diverse and include integrated assessments, life cycle assessments, interdependency assessments, and multi-criteria approaches. The main reason for this diversity is the multitude of sustainability dimensions ranging from environmental standards to corporate social responsibility. The diversity in methods brings challenges in terms of the interpretation and communication of outcomes. This claim is supported by De Olde et al. (2016), who find that producers hesitate to apply assessment tools because of the low degree of user-friendliness. Manfredi et al. (2018) support the demand for flexible methods to operationalize the assessment of agri-food sustainability. The comprehensive assessment framework developed in our research is based on an integrated approach, and it is tested for relevance and feasibility to facilitate the interpretation of outcomes.

### 6.2. Multi-Stage Analysis and Vertical and Horizontal Mechanisms

The literature on agri-food chain sustainability assessment mostly analyzes single-chain stages. The most evaluated is the production stage. The dominance of single-stage analyses can be explained by the complexity of assessing across different chain stages. For Biénabe et al. (2017) this complexity is due to geographical and sector-specific conditions. El Bilali (2019) also finds that legislation, policies, and standards at each supply chain stage complicate across-stage comparisons. Hassini et al. (2012) also claim that small and medium-sized enterprises are of limited concern to evaluators, which leads to key chain stages being ignored in assessment frameworks. Our comprehensive sustainability assessment framework covers multiple chain stages (from input supply to distribution and retail) and builds on a participatory approach. Future research could further explore the implications of integrating the consumption stage.

Including multiple supply chain stages in the analysis begs how these stages are interrelated and what this means for sustainability. Arshinder and Deshmukh (2008) argue that there is a lack of attention for the coordination dimension when assessing supply chain performance. Our structured literature review shows that the existing literature is still immature in defining the relationship between supply chain coordination and sustainability. Craven et al. (2016) find few connections between farmers' horizontal relations and economic performance and suggest more investigation on the relationship between farmers' ecological commitment and fair price mechanisms. We find agreement in the literature regarding the inclusion of coordination mechanisms being important in the multi-stage. The comprehensive sustainability assessment framework, therefore, includes both vertical and horizontal coordination indicators. These indicators support smallholders and small- and medium-sized enterprises in adopting sustainability practices.

Some studies within the sustainability assessment literature of agrifood chains seek to comprehend the connection between sustainability and chain coordination. Topics of these studies include the link between interdependencies and cooperation to reduce negative socio-economic or environmental impacts (Severo et al., 2018). Stakeholders committed to sustainable development could lean on the sustainabilityoriented coordination where responsibilities are clearly stated. Future research can further explore sustainability-oriented coordination to



Fig. 7. A conceptual framework for assessing sanustainability performance. Source: Authors' own representation



**Fig. 8.** Relevance of economic sustainability indicators. Source: Authors' own representation.



**Fig. 9.** Relevance of environmental sustainability indicators. Source: Authors' own representation.

derive the best practices. This may involve the inclusion of non-traditional actors, such as financial organizations or policymakers.

### 6.3. Architecture of the Comprehensive Framework

A broad range of appraisal methods responds to concerns recognised by the scientific community, chain actors, and policymakers. Concerns include the comprehensiveness of the framework and its robustness. Malak-Rawlikowska et al. (2019) and Roy and Chan (2012) strongly



**Fig. 10.** Relevance of social sustainability indicators. Source: Authors' own representation.

suggest that frameworks should move from a partial perspective towards an integrated and transdisciplinary approach (van der Ven, 2015; Sala et al. (2015). In this context, our assessment framework aims not to define an optimal framework but to define essential foundations. These specifications are joint sustainability dimensions, multi-stage chain analysis, vertical and horizontal mechanisms, and operationalization by applying an integrated assessment approach. The agri-food sustainability assessment literature suggests considering trade-offs between theory-based and empirical-based designs. This may also map out the steps forward to fulfill contextual perceptions that are vital in scientific research.

### 6.4. Practical Implications

The architecture of the framework enables us to advance towards transdisciplinarity. Fig. 6 represents the three foundations of the sustainability assessment framework. The first foundation, joint sustainability dimensions – economic, social, and environmental, is shown in green to indicate that this aspect is more frequently investigated. The second foundation is the multi-stage approach colored in yellow because it is less regularly incorporated. The third foundation is the coordination dimension composed of vertical and horizontal mechanisms and shown in red because it is an element of novelty in the sustainability assessment in agri-food chains.

The joint assessment of the sustainability indicators allows chain actors to evaluate trade-offs in performance on economic, environmental, and social dimensions. The weighting of the dimensions would



**Fig. 11.** a. Relevance of vertical coordination framework indicators. b. Relevance of horizontal coordination framework indicators. Source: Authors' own representation.

reflect context-specific needs and priorities. Trade-offs may exist across sustainability dimensions and should be incorporated in a comprehensive sustainability assessment.

The foundation of the multi-stage approach identifies the supply chain as the unit of analysis. It aims at incorporating information by actor, chain stage, linkage, or function. Scholars could apply this foundation to expand knowledge and distribute research benefits among Ecological Economics 192 (2022) 107265

chain actors. It also allows the framework to be used as a support tool. For instance, evidence of differences in sustainability performance between supply chains could be brought back to individual chain stages and may guide policymakers in supporting the development of specific chain stages or supply chains as a whole. The versatility of this foundation allows investigating single or multiple stages of supply chains. This may also aid supply chain actors and policymakers in indentifying the chain's priorities, regulative developments, and sustainable progress.

The foundation of coordination allows researchers and other stakeholders to explore the sustainability-coordination nexus further. Coordination indicators have the potential to cause a positive impact on the economic, environmental, and social performance at each chain stage and linkage. Chain actors can use information collected through this foundation to design strategies to foster competitiveness. For scholars and policymakers, it is essential to construct a solid contextualization of the chain before implementing the core foundations of the assessment framework. This contextualization should tackle political, ecological, social, technological, and legal aspects.

The results of the testing of the framework also lead to some practical implications. Relative relevance perceptions and feasibility assessment provide insights for practitioners about the most promising indicators to include in an assessment. The relevance allows connecting relevant indicators on the one hand with specific chain stages on the other hand. Feasibility results provide researchers and practitioners with guidance to quickly identify the most feasible indicators in fieldwork and evaluation. The study also provides the user with insights into assessment implementation. The review suggests to scholars and policymakers the aggregation of temporal scales by applying an ex-ante and ex-post evaluation and the cross-comparison of outcomes across supply chain stages.

Our research also has some limitations that scholars, policymakers, and chain actors should be aware of. First, the authors only considered peer-reviewed journals in their review. Significant knowledge may be found in other sources such as books, Ph.D. theses, and business papers. Second, even though the authors tried to employ appropriate search terms, some relevant terms may have been excluded (e.g., bioeconomy or circular economy). Third, representatives of environmental public institutions did not attend the participatory workshop; thus, the ecological view may have been underrepresented. Fourth, responses from fruits and vegetable stakeholders could portray only a part of the



**Fig. 12.** Feasibility of economic framework indicators. Source: Authors' own representation.



Fig. 13. Feasibility of environmental framework indicators. Source: Authors' own representation.



■ High ■ Medium ■ Low

Fig. 14. Feasibility of social framework indicators. Source: Authors' own representation.

sustainability challenges of the whole agri-food system since, for instance integrated supply chain would probably include more coordination and sustainable criteria.

### 7. Conclusions

This study performed a structured literature review of sustainability assessment frameworks in agri-food supply chains. Based on the structured literature review, we developed a comprehensive framework for measuring sustainability performance and tested this for relevance and feasibility. Four key findings follow from the study. First, the sustainability assessment framework developed in this article adds a relevant contribution to the existing literature on sustainability assessments. The proposed foundations of the framework operate as guidelines to

generate insights into sustainability assessment implementation. Second, comprehensiveness, user-friendliness, transdisciplinarity, and robustness are imperative aspects of framework development. Methods need testing and validation to boost the interpretation and communication of outcomes, as was done in the participatory workshop in this research. Third, when the studied context is the agri-food sector, the joint dimensions approach and the supply chain as a unit of analysis are the most suitable foundations. The joint dimensions approach allows for understanding the relationships between economic, environmental, and social sustainability aspects. As a unit of analysis, the supply chain allows comprehending how stages are interrelated and what this means for sustainability performance. Fourth, due to the nature of chain-stage connections, coordination mechanisms are key to developing organizational models to optimize economic utility, social welfare, and



Fig. 15. a. Feasibility of vertical coordination framework indicators. b. Feasibility of horizontal coordination framework indicators. Source: Authors' own representation.

ecological resilience.

### Funding

This research was financially supported by the Technical University of Ambato - Ecuador, throughout the agreement UTA-1887CU-P-2018, and by the European Union's Horizon 2020 research and innovation programme under grant agreement No 678024.

### **Declaration of Competing Interest**

This manuscript has not been published and is not under consideration for publication elsewhere. We have no conflicts of interest to disclose. The Technical University of Ambato provided the funding of the resources used to carry out the research. The funds used for in situ visits, information gathering, data collection, and interpretation of results. Finally, I would like to extend an **acknowledgment** to the Faculty of Food Science and Engineering of the Technical University of Ambato to support the development of this manuscript.

### Acknowledgments

The authors acknowledge support from representatives of the Development Agency CORPOAMBATO, and the Water Parliament and

Provincial Councils of Tungurahua.

### Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.ecolecon.2021.107265.

### References

- Abildtrup, J., Audsley, E., Fekete-Farkas, M., Giupponi, C., Gylling, M., Rosato, P., Rounsevell, M., 2006. Socio-economic scenario development for the assessment of climate change impacts on agricultural land use: a pairwise comparison approach. Environ. Sci. Policy 9, 101–115. https://doi.org/10.1016/j.envsci.2005.11.002.
- Ansari, Z.N., Kant, R., 2017. A state-of-art literature review reflecting 15 years of focus on sustainable supply chain management. J. Clean. Prod. 142, 2524–2543. https:// doi.org/10.1016/j.jclepro.2016.11.023.
- Aramyan, L., Ondersteijn, J.M., Kooten, O., Oude-Lansink, A., 2006. Performance indicators in agri-food production chains. In: Quantifying the Agri-Food supply Chain, 5, pp. 49–66. https://doi.org/10.1007/1-4020-4693-6\_5.
- Arshinder, K.A., Deshmukh, S.G., 2008. Supply chain coordination: perspectives, empirical studies and research directions. Int. J. Prod. Econ. 115 (2), 316–335. https://econpapers.repec.org/article/eeeproeco/v\_3a115\_3ay\_3a2008\_3ai\_3a2\_3ap\_3a31 6-335.htm.
- Barbosa, M., 2021. Uncovering research streams on agri-food supply chain management: a bibliometric study. Global Food Security. 28, 05–17. https://doi.org/10.1016/j. gfs.2021.100517.
- Baudino, C., Giuggioli, N.R., Briano, R., Massaglia, S., Peano, C., 2017. Integrated methodologies (SWOT, TOWS, LCA) for improving production chains and environmental sustainability of kiwifruit and baby kiwi in Italy. Sustainability. 9, 1621. https://doi.org/10.3390/su9091621.

Beske, P., Seuring, S., 2014. Putting sustainability into supply chain management. Supply Chain Manag. 19, 322–331. https://doi.org/10.1108/SCM-12-2013-0432.

Biénabe, E., Rival, A., Loeillet, D., 2017. Sustainable Development and Tropical Agri-Chains. Springer Netherlands. https://doi.org/10.1007/978-94-024-1016-7.

- Bijman, J., Omta, S.W.F., Trienekens, J.H., Wijnands, J.H.M., Wubben, E.F.M., 2006. International Agri-Food Chains and Networks: Management and Organization, second ed. Wageningen Academic Publishers, Wageningen. https://doi.org/ 10.3920/978-90-8686-573-4.
- Boesen, S., Bey, N., Niero, M., 2019. Environmental sustainability of liquid food packaging: is there a gap between Danish consumers' perception and learnings from life cycle assessment? J. Clean. Prod. 210, 1193–1206. https://doi.org/10.1016/j. jclepro.2018.11.055.
- Borgen, S.O., Hegrenes, A., 2005. How Can Transaction Cost Economics Add to the Understanding of New Contractual Formats in the Norwegian Agri-food System? (No. 82-7077-608–4).
- Camargo, M.C., Hogarth, N.J., Pacheco, P., Nhantumbo, I., Kanninen, M., 2019. Greening the dark side of chocolate: a qualitative assessment to inform sustainable supply chains. Environ. Conserv. 46, 9–16. https://doi.org/10.1017/S0376892918000243.
- Carter, C.R., Easton, P.L., 2011. Sustainable supply chain management: evolution and future directions. Int. J. Phys. Distrib. Logist. Manag. 41, 46–62. https://doi.org/ 10.1108/09600031111101420.
- Carter, C.R., Rogers, D.S., 2008. A framework of sustainable supply chain management: moving toward new theory. Int. J. Phys. Distrib. Logist. Manag. 38, 360–387. https://doi.org/10.1108/09600030810882816.
- Chaparro, A., Calle, Á., 2017. Peasant economy sustainability in peasant markets, Colombia. Agroecol. Sustain. Food Syst. 41, 204–225. https://doi.org/10.1080/ 21683565.2016.1266069.
- Charles, R., Jolliet, O., Gaillard, G., Pellet, D., 2006. Environmental analysis of intensity level in wheat crop production using life cycle assessment. Agric. Ecosyst. Environ. 113, 216–225. https://doi.org/10.1016/j.agee.2005.09.014.
- Chen, L., Zhao, X., Tang, O., Price, L., Zhang, S., Zhu, W., 2017. Supply chain collaboration for sustainability: a literature review and future research agenda. Int. J. Prod. Econ. 194, 73–87. https://doi.org/10.1016/j.ijpe.2017.04.005.
- Cornelissen, A.M.G., Van Den Berg, J., Koops, W.J., Grossman, M., Udo, H.M.J., 2001. Assessment of the contribution of sustainability indicators to sustainable development: a novel approach using fuzzy set theory. Agric. Ecosyst. Environ. 86, 173–185. https://doi.org/10.1016/S0167-8809(00)00272-3.
- Craven, T., Mittal, A., Krejci, C., 2016. Effective coordination in regional food supply chain. Indust. Manufact. Syst. Eng. 104. Retrieved from. http://lib.dr.iastate. edu/imse conf/104.
- Cronin, P., Ryan, F., Coughlan, M., 2008. Undertaking a literature review: a step-by-step approach. Brit. J. Nurs 17, 38–43. https://doi.org/10.12968/bjon.2008.17.1.28059.
- Dania, W.A.P., Xing, K., Amer, Y., 2018. Collaboration behavioural factors for sustainable agri-food supply chains: a systematic review. J. Clean. Prod. 186, 851–864. https://doi.org/10.1016/j.jclepro.2018.03.148.
- Dasgupta, P., 1996. The economics of the environment. Environ. Dev. Econ. 1 (4). Retrieved from. https://www.jstor.org/stable/44379225.
- De Olde, E.M., Oudshoorn, F.W., Sørensen, C.A.G., Bokkers, E.A.M., De Boer, I.J.M., 2016. Assessing sustainability at farm-level: lessons learned from a comparison of tools in practice. Ecol. Indic. 66, 391–404. https://doi.org/10.1016/j. ecolind.2016.01.047.
- Diener, E., Suh, E., 1997. Measuring quality of life: economic, social, and subjective indicators. Soc. Indic. Res. 40, 189–216. https://doi.org/10.1023/A: 1006859511756.
- Douma, S., Schreuder, H., 2017. Economic Approaches to Organization, sixth ed. Pearson Publishing, New York. https://doi.org/10.1177/017084069301400213.
- Dreyer, L.C., Hauschild, M.Z., Schierbeck, J., 2006. A framework for social life cycle impact assessment. Int. J. Life Cycle Assess. 11, 88–97. https://doi.org/10.1065/ lca2005.08.223.
- El Bilali, H., 2019. The multi-level perspective in research on sustainability transitions in agriculture and food systems: a systematic review. Agriculture (Switzerland). 9, 74–88. https://doi.org/10.3390/agriculture9040074.
- Engert, S., Baumgartner, R.J., 2016. Corporate sustainability strategy bridging the gap between formulation and implementation. J. Clean. Prod. 113, 822–834. https://doi. org/10.1016/j.jclepro.2015.11.094.
- Estevez, C.L., Bhat, M.G., Bray, D.B., 2018. Commodity chains, institutions, and domestic policies of organic and fair trade coffee in Bolivia. Agroecol. Sustain. Food Syst. 42, 299–327. https://doi.org/10.1080/21683565.2017.1359737.
- Fan, M.N., Leu, J. Der, Krischke, A., 2016. The evaluation of green manufacturing: A DEA-based approach. In: IEEE International Conference on Industrial Engineering and Engineering Management, 2016-December, pp. 65–69. https://doi.org/ 10.1109/IEEM.2016.7797837.
- Farooque, M., Zhang, A., Thürer, M., Qu, T., Huisingh, D., 2019. Circular supply chain management: a definition and structured literature review. J. Clean. Prod. 228, 882–900. https://doi.org/10.1016/j.jclepro.2019.04.303.
- Forman, E.H., Gass, S.I., 2001. The analytic hierarchy process—an exposition. Oper. Res. 49, 469–486. https://doi.org/10.1287/opre.49.4.469.11231.
- Formentini, M., Taticchi, P., 2016. Corporate sustainability approaches and governance mechanisms in sustainable supply chain management. J. Clean. Prod. 112, 1920–1933. https://doi.org/10.1016/j.jclepro.2014.12.072.
- Gerbens-Leenes, P.W., Moll, H.C., Schoot Uiterkamp, A.J.M., 2003. Design and development of a measuring method for environmental sustainability in food production systems. Ecol. Econ. 46, 231–248. https://doi.org/10.1016/S0921-8009 (03)00140-X.

- Gereffi, G., 1994. The Organization of Buyer-Driven Global Commodity Chains. In Commodity Chains and Global Capitalism. Praeger Publishers, California. https:// doi.org/10.1177/102452949600100406.
- Gereffi, G., Humphrey, J., Sturgeon, T., 2005. The governance of global value chains. Rev. Int. Polit. Econ. 12, 78–104. https://doi.org/10.1080/09692290500049805.
- Gimenez, C., Tachizawa, E.M., 2012. Extending sustainability to suppliers: a systematic literature review. Supply Chain Manag. 17, 531–543. https://doi.org/10.1108/ 13598541211258591.
- Godde, C.M., Mason-D'Croz, D., Mayberry, D.E., Thornton, P.K., Herrero, M., 2021. Impacts of climate change on the livestock food supply chain; a review of the evidence. Global Food Security. 28, 04–88. https://doi.org/10.1016/j. gfs.2020.100488.
- Golicic, S.L., Smith, C.D., 2013. A meta-analysis of environmentally sustainable supply chain management practices and firm performance. J. Supply Chain Manag. 49, 78–95. https://doi.org/10.1111/jscm.12006.
- Gonzalez-Perez, M.A., Gutierrez-Viana, S., 2012. Cooperation in coffee markets: the case of Vietnam and Colombia. J. Agribus. Dev. Emerg. Econ. 2, 57–73. https://doi.org/ 10.1108/20440831211219237.
- Gouda, S.K., Saranga, H., 2018. Sustainable supply chains for supply chain sustainability: impact of sustainability efforts on supply chain risk. Int. J. Prod. Res. 56, 5820–5835. https://doi.org/10.1080/00207543.2018.1456695.
- Govindan, K., 2018. Sustainable consumption and production in the food supply chain: a conceptual framework. Int. J. Prod. Econ. 195, 419–431. https://doi.org/10.1016/j. ijpe.2017.03.003.
- Green, A., Nemecek, T., Chaudhary, A., Mathys, A., 2020. Assessing nutritional, health, and environmental sustainability dimensions of agri-food production. Global Food Security. 26, 04–062. https://doi.org/10.1016/j.gfs.2020.100406.
- Gregory, A.J., Atkins, J.P., Midgley, G., Hodgson, A.M., 2020. Stakeholder identification and engagement in problem structuring interventions. Eur. J. Oper. Res. 283, 321–340. https://doi.org/10.1016/j.ejor.2019.10.044.
- Grimm, J.H., Hofstetter, J.S., Sarkis, J., 2016. Exploring sub-suppliers' compliance with corporate sustainability standards. J. Clean. Prod. 112, 1971–1984. https://doi.org/ 10.1016/j.jclepro.2014.11.036.
- Gunawan, A.A., van Riel, A., Essers, C., 2021. What drives ecopreneurship in women and men? - A structured literature review. J. Clean. Prod. 280, 124–336. https://doi.org/ 10.1016/j.jclepro.2020.124336.
- Hasna, A., 2009. A review of sustainability assessment methods in engineering. Int. J. Environ. Cult. Econ. Social Sustain 5, 20–39. https://doi.org/10.18848/1832-2077/ CGP/v05i01/54552.
- Hassini, E., Surti, C., Searcy, C., 2012. A literature review and a case study of sustainable supply chains with a focus on metrics. Int. J. Prod. Econ. 140 (1), 69–82. https://doi. org/10.1016/j.ijpe.2012.01.042.
- Herzog, A., Winckler, C., Zollitsch, W., 2018. In pursuit of sustainability in dairy farming: a review of interdependent effects of animal welfare improvement and environmental impact mitigation. Agric. Ecosyst. Environ. 267, 174–187. https:// doi.org/10.1016/j.agee.2018.07.029.
- Hoekstra, A.Y., Chapagain, A.K., van Oel, P.R., 2019. Progress in water footprint assessment: towards collective action in water governance. Water (Switzerland). 11, 13–26. https://doi.org/10.3390/w11051070.
- Holmberg, S., 2000. A systems perspective on supply chain measurements. Int. J. Phys. Distrib. Logistics Manage. 30, 847–868. https://doi.org/10.1108/ 09600030010351246.
- Huang, Y., Wang, K., Zhang, T., Pang, C., 2016. Green supply chain coordination with greenhouse gases emissions management: a game-theoretic approach. J. Clean. Prod. 112, 2004–2014. https://doi.org/10.1016/j.jclepro.2015.05.137.
- Huss, M., Brander, M., Kassie, M., Ehlert, U., Bernauer, T., 2021. Improved storage mitigates vulnerability to food-supply shocks in smallholder agriculture during the COVID-19 pandemic. Global Food Security. 28, 04–68. https://doi.org/10.1016/j. gfs.2020.100468.
- Ivanov, D., Dolgui, A., 2020. Viability of intertwined supply networks: extending the supply chain resilience angles towards survivability. A position paper motivated by COVID-19 outbreak. Int. J. Prod. Res. 58, 2904–2915. https://doi.org/10.1080/ 00207543.2020.1750727.
- Jiao, Y., Su, M., Ji, C., Yang, S., Zhang, P., 2021. How to design fully cooperative policies to abate transboundary air pollution between two highly asymmetric regions: an abnormal incrementalism analysis. J. Clean. Prod. 278 https://doi.org/10.1016/j. jclepro.2020.124042.
- Kilelu, C., Klerkx, L., Omore, A., Baltenweck, I., Leeuwis, C., Githinji, J., 2017. Value chain upgrading and the inclusion of smallholders in markets: reflections on contributions of multi-stakeholder processes in dairy development in Tanzania. Eur. J. Dev. Res. 29, 1102–1121. https://doi.org/10.1057/s41287-016-0074-z.
- Koberg, E., Longoni, A., 2019. A systematic review of sustainable supply chain management in global supply chains. J. Clean. Prod. 207, 1084–1098. https://doi. org/10.1016/j.jclepro.2018.10.033.
- Kornher, L., Kalkuhl, M., 2019. The gains of coordination when does regional cooperation for food security make sense? Global Food Security. 22, 37–45. https:// doi.org/10.1016/j.gfs.2019.09.004.
- Kozlowski, A., Searcy, C., Bardecki, M., 2015. Corporate sustainability reporting in the apparel industry. Int. J. Product. Perform. Manag. 64, 377–397. https://doi.org/ 10.1108/IJPPM-10-2014-0152.
- Kugelberg, S., Bartolini, F., Kanter, D.R., Milford, A.B., Pira, K., Sanz-Cobena, A., Leip, A., 2021. Implications of a food system approach for policy agenda-setting design. Global Food Security. 28, 04–51. https://doi.org/10.1016/j.gfs.2020.100451.
- Lee, S.Y., Hu, J., Lim, M.K., 2021. Maximising the circular economy and sustainability outcomes: an end-of-life Tyre recycling outlets selection model. Int. J. Prod. Econ. 232, 107–965. https://doi.org/10.1016/j.ijpe.2020.107965.

### C. Moreno-Miranda and L. Dries

Lejars, C., Fusillier, J.L., Bouarfa, S., Coutant, C., Brunel, L., Rucheton, G., 2012. Limitation of agricultural ground water uses in Beauce (France): what are the impacts on farms and on the food-processing sector? Irrig. Drain. 61, 54–64. https:// doi.org/10.1002/IRD.1659.

- Lie, H., Rich, K.M., Kurwijila, L.R., Jervell, A.M., 2012. Improving smallholder livelihoods through local value chain development: a case study of goat milk yogurt in Tanzania. Int. Food Agribus. Manage. Rev. 15, 20–37. https://www.ifama.org/re sources/Documents/v15i3/Lie-Rich-Jervell.pdf.
- Longo, S., Mistretta, M., Guarino, F., Cellura, M., 2017. Life cycle assessment of organic and conventional apple supply chains in the north of Italy. J. Clean. Prod. 140, 654–663. https://doi.org/10.1016/j.jclepro.2016.02.049.
- Luo, L., van der Voet, E., Huppes, G., 2009. Life cycle assessment and life cycle costing of bioethanol from sugarcane in Brazil. Renew. Sust. Energ. Rev. 13, 1613–1619. https://doi.org/10.1016/j.rser.2008.09.024.
- Macdonald, K., 2020. Private sustainability standards as tools for empowering southern pro-regulatory coalitions? Collaboration, conflict and the pursuit of sustainable palm oil. Ecol. Econ. 167, 10–64. https://doi.org/10.1016/J.ECOLECON.2019.106439.
- Malak-Rawlikowska, A., Majewski, E., Was, A., Borgen, S.O., Csillag, P., Donati, M., Wavresky, P., 2019. Measuring the economic, environmental, and social sustainability of short food supply chains. Sustainability (Switzerland) 11, 4004. https://doi.org/10.3390/su11154004.
- Malone, T.W., Crowston, K., 1994. The interdisciplinary study of coordination. ACM Comput. Surv. (CSUR). 26, 87–119. https://doi.org/10.1145/174666.174668.
- Manfredi, A.P., Ballesteros, I., Sáez, F., Perotti, N.I., Martínez, M.A., Negro, M.J., 2018. Integral process assessment of sugarcane agricultural crop residues conversion to ethanol. Bioresour. Technol. 260, 241–247. https://doi.org/10.1016/j. biortech.2018.03.114.
- Mani, V., Jabbour, C.J.C., Mani, K., 2020. Supply chain social sustainability in small and medium manufacturing enterprises and firms' performance: empirical evidence from an emerging Asian economy. Int. J. Prod. Econ. 227, 107656. https://doi.org/ 10.1016/i.ijpe.2020.107656.
- Manna, Z., Waldinger, R., 1986. A deductive approach to program synthesis. Read. Artif. Intel. Softw. Eng. 1, 3–34. https://doi.org/10.1016/b978-0-934613-12-5.50006-9.
- Martins, C.L., Pato, M.V., 2019. Supply chain sustainability: a tertiary literature review. J. Clean. Prod. 225, 995–1016. https://doi.org/10.1016/j.jclepro.2019.03.250.
- Mausch, K., Hall, A., Hambloch, C., 2020. Colliding paradigms and trade-offs: agri-food systems and value chain interventions. Global Food Security. 26, 04–39. https://doi. org/10.1016/j.gfs.2020.100439.
- Melkonyan, A., Gruchmann, T., Lohmar, F., Kamath, V., Spinler, S., 2020. Sustainability assessment of last-mile logistics and distribution strategies: the case of local food networks. Int. J. Prod. Econ. 228, 10–46. https://doi.org/10.1016/j. iipe.2020.107746.
- Mihai, F.C., Ingrao, C., 2018. Assessment of biowaste losses through unsound waste management practices in rural areas and the role of home composting. J. Clean. Prod. 172, 1631–1638. https://doi.org/10.1016/j.jclepro.2016.10.163.
- Moreno-Miranda, C., Palacios, H., Rama, D., 2019. Small-holders perception of sustainability and chain coordination: evidence from Arriba PDO Cocoa in Western Ecuador. Bio-Based Appl. Econ. 8, 279–295. https://doi.org/10.13128/BAE-9448.
- Moreno-Miranda, C., Paredes, M.F., Solís, N., Moreno, R., Rama, D., 2020. Structural analysis of nontraditional Andean fruit chains: the case of the Inca berry agri-food network in Ecuador. J. Agric. Environ. Int. Dev. 114, 57–76. https://doi.org/ 10.12895/JAEID.20201.934.
- Moustier, P., Leplaideur, A., 1999. Cadre d'analyse des acteurs du commerce vivrier africain. CIRAD, Montpellier.
- Msaddak, M., Bennasr, J., Zaibet, L., Fridhi, M., 2017. Social networks for the sustainability of the dairy sector: the role of cooperatives. Livest. Res. Rural. Dev. 29, 50–77. http://www.lrrd.org/lrrd29/2/msad29028.html.
- Mutebi Kalibwani, R., Twebaze, J., Kamugisha, R., Kakuru, M., Sabiiti, M., Kugonza, I., Nyamwaro, S., 2018. Multi-stakeholder partnerships in value chain development: a case of the organic pineapple in Ntungamo district, Western Uganda. J. Agribus. Dev. Emerg. Econ. 8, 171–185. https://doi.org/10.1108/JADEE-08-2015-0038.
- Neves, M.F., 2010. Inserting small holders into sustainable value chains. In: Handbook of Business Practices and Growth in Emerging Markets, 12, pp. 235–254. https://doi. org/10.1142/9789812791788\_0013.
- Niemeijer, D., de Groot, R.S., 2008. A conceptual framework for selecting environmental indicator sets. Ecol. Indic. 8, 14–25. https://doi.org/10.1016/j.ecolind.2006.11.012. Paciarotti, C., Torregiani, F., 2021. The logistics of the short food supply chain: a
- Paciarotti, C., Torregiani, F., 2021. The logistics of the short food supply chain: a literature review. Sustain. Prod. Consump. 26, 428–442. https://doi.org/10.1016/j. spc.2020.10.002.
- Panagiotakopoulos, P.D., Espinosa, A., Walker, J., 2016. Sustainability management: insights from the viable system model. J. Clean. Prod. 113, 792–806. https://doi. org/10.1016/j.jclepro.2015.11.035.
- Peterson, H., 2010. Transformational supply chains and the "wicked problem" of sustainability: aligning knowledge, innovation, entrepreneurship, and leadership. J. Chain Network Sci. 9, 71–82. https://doi.org/10.3920/jcns2009.x178.
- Peterson, H.C., Wysocki, A., Harsh, S.B., 2001. Strategic choice along the vertical coordination continuum. Int. Food Agribus. Manage. Rev. 4, 149–166. https://doi. org/10.1016/S1096-7508(01)00079-9.
- Porter, M., 1985. Competitive Advantage: Creating and Sustaining Superior Performance. Simon and Schuster, New York. https://doi.org/10.1057/978-1-137-00772-8\_649.
- Pullman, M.E., Maloni, M.J., Dillard, J., 2010. Sustainability practices in food supply chains: how is wine different? J. Wine Res. 21, 35–56. https://doi.org/10.1080/ 09571264.2010.495853.

- Rao, P., Holt, D., 2005. Do green supply chains lead to competitiveness and economic performance? Int. J. Operat. Prod. Manage. 25, 898–916. https://doi.org/10.1108/ 01443570510613956.
- Raynold, L., 2004. The globalization of organic agro-food networks. World Dev. 32, 725–743.
- Roy, R., Chan, N.W., 2012. An assessment of agricultural sustainability indicators in Bangladesh: review and synthesis. Environmentalist. 32, 99–110. https://doi.org/ 10.1007/s10669-011-9364-3.
- Rydberg, T., Haden, A.C., 2006. Emergy evaluations of Denmark and Danish agriculture: assessing the influence of changing resource availability on the organization of agriculture and society. Agric. Ecosyst. Environ. 117, 145–158. https://doi.org/ 10.1016/j.agee.2006.03.025.
- Sala, S., Ciuffo, B., Nijkamp, P., 2015. A systemic framework for sustainability assessment. Ecol. Econ. 119, 314–325. https://doi.org/10.1016/j. ecolecon.2015.09.015.
- Schanes, K., Dobernig, K., Gözet, B., 2018. Food waste matters a systematic review of household food waste practices and their policy implications. J. Clean. Prod. 182, 978–991. https://doi.org/10.1016/j.jclepro.2018.02.030.
- Schmidt, X., Espinoza, N., Azapagic, A., 2014. Life cycle environmental impacts of convenience food: comparison of ready and home-made meals. J. Clean. Prod. 73, 294–309. https://doi.org/10.1016/j.jclepro.2014.01.008.
- Seuring, S., Gold, S., 2012. Conducting content-analysis based literature reviews in supply chain management. Supply Chain Manage. Int. J. 17, 544–555. https://doi. org/10.1108/13598541211258609.
- Severo, E.A., de Guimarães, J.C.F., Henri, E.C., 2018. Cleaner production, social responsibility and eco-innovation: generations' perception for a sustainable future. J. Clean. Prod. 186, 91–103. https://doi.org/10.1016/j.jclepro.2018.03.129.
- Shafiq, A., Klassen, R.D., Johnson, P.F., Awaysheh, A., 2014. Socially responsible practices: an exploratory study on scale development using stakeholder theory. Decis. Sci. 45, 683–716. https://doi.org/10.1111/deci.12085.
- Slange, L., Loucks, L., Slangen, A., 2008. Institutional economics and economic organisation theory. Wageningen Academic Publishers, Wageningen. https://doi. org/10.1093/erae/jbp015.
- Small, H., 1980. Co-citation context analysis and the structure of paradigms. J. Doc. 36, 183–196. https://doi.org/10.1108/eb026695.
- Sting, F.J., Loch, C.H., 2016. Implementing operations strategy: how vertical and horizontal coordination interact. Prod. Oper. Manag. 25, 1177–1193. https://doi. org/10.1111/poms.12537.
- Sutton, A.J., Jones, D.R., Abrams, K.R., Sheldon, T.A., Song, F., 1999. Systematic reviews and meta-analysis: a structured review of the methodological literature. J. Health Serv. Res. Policy. 4, 49–55. https://doi.org/10.1177/135581969900400112.
- Tachizawa, E.M., Wong, C.Y., 2014. Towards a theory of multi-tier sustainable supply chains: a systematic literature review. Supply Chain Manag. 19, 643–653. https:// doi.org/10.1108/SCM-02-2014-0070.
- Thomas, D.R., 2006. A general inductive approach for analyzing qualitative evaluation data. Am. J. Eval. 27, 237–246. https://doi.org/10.1177/1098214005283748.
- Thomassen, M.A., Dolman, M.A., van Calker, K.J., de Boer, I.J.M., 2009. Relating life cycle assessment indicators to gross value added for Dutch dairy farms. Ecol. Econ. 68, 2278–2284. https://doi.org/10.1016/j.ecolecon.2009.02.011.
   Tost, M., Hitch, M., Chandurkar, V., Moser, P., Feiel, S., 2018. The state of environmental
- Tost, M., Hitch, M., Chandurkar, V., Moser, P., Feiel, S., 2018. The state of environmental sustainability considerations in mining. J. Clean. Prod. 182, 969–977. https://doi. org/10.1016/j.jclepro.2018.02.051.
- Um, K.H., Kim, S., 2019. The effects of supply chain collaboration on performance and transaction cost advantage: the moderation and nonlinear effects of governance mechanisms. Int. J. Prod. Econ. 217, 97–111. https://doi.org/10.1016/j. iipe.2018.03.025.
- Van Čauwenbergh, N., Biala, K., Bielders, C., Brouckaert, V., Franchois, L., Garcia Cidad, V., Hermy, M., Mathijs, E., Muys, B., Reijnders, J., Sauvenier, X., Valckx, J., Vanclooster, M., Van der Veken, B., Wauters, E., Peeters, A., 2007. SAFE-A hierarchical framework for assessing the sustainability of agricultural systems. Agric. Ecosyst. Environ. 120, 229–242. https://doi.org/10.1016/j.agee.2006.09.006.
- van der Ven, H., 2015. Correlates of rigorous and credible transnational governance: a cross-sectoral analysis of best practice compliance in eco-labeling. Regul. Govern. 9, 276–293. https://doi.org/10.1111/rego.12092.
- Van Der Werf, H.M.G., Petit, J., 2002. Evaluation of the environmental impact of agriculture at the farm level: a comparison and analysis of 12 indicator-based methods. Agric. Ecosyst. Environ. 93, 131–145. https://doi.org/10.1016/S0167-8809(01)00354-1.
- Vanlauwe, B., Hungria, M., Kanampiu, F., Giller, K.E., 2019. The role of legumes in the sustainable intensification of African smallholder agriculture: lessons learnt and challenges for the future. Agric. Ecosyst. Environ 284, 106–583. https://doi.org/ 10.1016/j.agee.2019.106583.
- Von Wirén-Lehr, S., 2001. Sustainability in agriculture an evaluation of principal goaloriented concepts to close the gap between theory and practice. Agric. Ecosyst. Environ. 84, 115–129. https://doi.org/10.1016/S0167-8809(00)00197-3.
- Vroegindewey, R., Theriault, V., Staatz, J., 2018. Coordinating cereal farmers and buyers: evidence from Mali. J. Agribus. Dev. Emerg. Econ. 8, 234–255. https://doi. org/10.1108/JADEE-11-2016-0075.
- Wang, Y., Luo, W., Zeng, G., Peng, H., Cheng, A., Zhang, L., Cai, X., Chen, J., Lyu, Y., Yang, H., Wang, S., 2020. Characteristics of carbon, water, and energy fluxes on abandoned farmland revealed by critical zone observation in the karst region of southwest China. Agric. Ecosyst. Environ. 292, 106–821. https://doi.org/10.1016/j. agee.2020.106821.
- Ward, P.S., Bell, A.R., Parkhurst, G.M., Droppelmann, K., Mapemba, L., 2016. Heterogeneous preferences and the effects of incentives in promoting conservation

### C. Moreno-Miranda and L. Dries

agriculture in Malawi. Agric. Ecosyst. Environ. 222, 67-79. https://doi.org/ 10.1016/j.agee.2016.02.005.

- Webster, J., Watson, R.T., 2002. Analyzing the past to prepare for the future: writing a
- literature review. MIS Quart. 26, 78–95. https://doi.org/10.2307/4132319. Wustenberghs, H., Coteur, I., Debruyne, L., Marchand, F., 2015. Survey of Sustainability Assessment Methods. http://tempag.net/documents/survey-of-sustainability-asse ssment-methods/ accessed 18 August 2020.
- Xie, G., 2016. Cooperative strategies for sustainability in a decentralized supply chain with competing suppliers. J. Clean. Prod. 113, 807-821. https://doi.org/10.1016/j. jclepro.2015.11.013.
- Yawar, S.A., Seuring, S., 2017. Management of social issues in supply chains: a literature review exploring social issues, actions and performance outcomes. J. Busin. Ethics. 141, 621–643. https://doi.org/10.1007/s10551-015-2719-9. Zhong, Z., Zhang, C., Jia, F., Bijman, J., 2018. Vertical coordination and cooperative
- member benefits: case studies of four dairy farmers' cooperatives in China. J. Clean. Prod. 172, 2266-2277. https://doi.org/10.1016/j.jclepro.2017.11.184.
- Zhu, Q., Feng, Y., Choi, S.B., 2017. The role of customer relational governance in environmental and economic performance improvement through green supply chain management. J. Clean. Prod. 155, 46-53. https://doi.org/10.1016/j. jclepro.2016.02.124.